Operability Report 

12/13/2006
Accelerator
Maintain 5 pass 3.303 GeV machine (326 MeV per linac)
· High current run for A and C lasers
· Weekly Mott measurements
· No beam to BSY dump
· New Orbit locks for halls A and C (To replace PID locks)

Hall A

Elastic Electron Scattering Off 3He and 4He
· 5 pass 3.303 GeV
· Change to 2 pass sometime next week (1 day of running at 1 pass w/hall C off)???
· Up to 100 uA

· No beam in Compton chicane
· FFB on position and energy
· ep or arc energy measurement as requested (procedure will be updated)
Hall B

G 13 Kaon Production on the Deuteron Using Polarized Photons
· 4 pass 2.649 GeV
· 1 to 40 nA
· Current and orbit locks on

· 60 Hz noise vs aperture steering

· Development of 60 Hz suppression tool in progress

Hall C

G0 Backwards Angle







· 1 pass 0.690 GeV
· Up to 40 uA

· Weekly Mott, Moeller, and arc energy measurements
· RF phase vs halo

· Position asymmetry distribution (Riad test plan today to address this)
· FFB on position only
· Vertical motion downstream of Moeller region (3C17A) eliminated with new Moeller Off optics installed 
· Frequent iocse20 reboots (G0 BPM’s), Radiation induced. 
· Turned off Moeller quads, installed new hall C target orbit lock, stabilized vertical position from 3C17-3HG0.  
Items Needing Attention
Beam Transport Issues 

Accelerator
· BLM trips at high current (AT07, 8E02, 1L02, 7R03) 
· Scraping at 1st pass extraction point. 

· Indicated position at 2E02 BPM was ~3mm off.  Centering in horizontal aperture eliminated BLM activity and allowed hall A to run at max current (85uA) 
· 1A SLM instability has not been isolated to the injector vs. NL RF yet.

· Search by R Kazimi, C Mounts, and B Team is inconclusive. 

Hall A

· Accelerator BLM trips at high current target. 
Hall B

· Current stability, 60 Hz noise, Charge Asymmetry
· Beam centered on hall B chopping slit, Path length adjusted
Hall C






· Combined optics solution is still a possibility
· High halo on 10/31/06 caused by DC beam coming from the hall A laser.
· High halo on 11/2/06 caused by stray magnetic field after Moeller back out
· Adjusted quad at 3C16 to lower rates 
